Perinatal mental health services in pregnancy and the year after birth: the ESMI research programme including RCT

Louise M Howard,^{1*} Kathryn M Abel,²
Katie H Atmore,¹ Debra Bick,³ Amanda Bye,¹
Sarah Byford,⁴ Lauren E Carson,¹ Clare Dolman,¹
Margaret Heslin,⁴ Myra Hunter,⁵ Stacey Jennings,¹
Sonia Johnson,⁶ Ian Jones,⁷ Billie Lever Taylor,⁶
Rebecca McDonald,¹ Jeannette Milgrom,⁸
Nicola Morant,⁶ Selina Nath,¹ Susan Pawlby,⁹
Laura Potts,¹⁰ Claire Powell,¹ Diana Rose,¹¹
Elizabeth Ryan,¹⁰ Gertrude Seneviratne,¹²
Rebekah Shallcross,¹ Nicky Stanley,¹³
Kylee Trevillion,¹ Angelika Wieck¹⁴ and
Andrew Pickles¹⁰

¹Section of Women's Mental Health, Institute of Psychiatry, Psychology and Neuroscience, King's College London, London, UK

²Centre for Women's Mental Health, The University of Manchester, Manchester, UK

³Division of Women and Children's Health, Florence Nightingale Faculty of Nursing, Midwifery and Palliative Care, King's College London, London, UK

⁴King's Health Economics, Institute of Psychiatry, Psychology and Neuroscience, King's College London, London, UK

⁵Department of Psychology, Institute of Psychiatry, Psychology and Neuroscience, King's College London, London, UK

⁶Division of Psychiatry, University College London, London, UK

⁷Institute of Psychological Medicine and Clinical Neuroscience, Cardiff University, Cardiff, UK

⁸Department of Clinical and Health Psychology, Parent-Infant Research Institute, University of Melbourne, Melbourne, VIC, Australia

⁹Department of Psychological Medicine, Institute of Psychiatry, Psychology and Neuroscience, King's College London, London, UK

¹⁰Biostatistics and Health Informatics, King's College London, London, UK

¹¹Service User Research Enterprise, Institute of Psychiatry, Psychology and Neuroscience, King's College London, London, UK

¹²Perinatal Service, South London and Maudsley NHS Foundation Trust, London, UK

¹³School of Social Work, Care and Community, University of Central Lancashire, Harrington, UK

¹⁴Greater Manchester Mental Health NHS Foundation Trust, Manchester, UK

^{*}Corresponding author louise.howard@kcl.ac.uk

Declared competing interests of authors: Louise M Howard chaired the National Institute for Health and Care Excellence (NICE) Antenatal and Postnatal Mental Health Guidance [Clinical Guidance 192 (CG192); 2014], has worked for NICE Scientific Advice (London, UK) and was partly funded (2018-19) by the National Institute for Health and Care Research (NIHR) Biomedical Research at the South London and Maudsley NHS Foundation Trust (London, UK) and King's College London (London, UK). Louise M Howard was also supported by a NIHR Research Professorship (NIHR-RP-R3-12-011). Debra Bick was supported by the NIHR Collaboration for Leadership in Applied Health Research and Care South London at King's College Hospital NHS Foundation Trust (London, UK) and is an active member of the NIHR Health Services and Delivery Research (HSDR) Funding Committee (2020 to present). Ian Jones reports grants from Takeda Pharmaceutical Company Limited (Tokyo, Japan) and personal fees from Sanofi SA (Paris, France), outside the submitted work. In addition, Ian Jones is a trustee of Action on Postpartum Psychosis (Swansea, UK) and is a trustee of the Maternal Mental Health Alliance (London, UK). Rebecca McDonald reports personal fees from Improving Outcomes in the Treatment of Opioid Dependence (London, UK), other funding from Mundipharma Research Ltd (Cambridge, UK) and personal fees from the United Nations Office on Drug Crime (Vienna, Austria), outside the submitted work. Elizabeth Ryan is funded by the Medical Research Council (London, UK; grant MR/N028287/1). Rebekah Shallcross is funded by a NIHR HSDR grant (16/117/03) and grants from the Wellcome Institutional Strategic Support Fund Return to Work Fellowship, outside the submitted work. Kylee Trevillion is funded by a NIHR HSDR grant (16/117/03). Andrew Pickles received support from the NIHR King's College London/South London and Maudsley NHS Foundation Trust Biomedical Research Centre (IS-BRC-1215-20018).

Published June 2022 DOI: 10.3310/CCHT9881

Plain English summary

The ESMI research programme including RCT Programme Grants for Applied Research 2022; Vol. 10: No. 5 DOI: 10.3310/CCHT9881

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

E arly identification and effective treatment of mothers with mental illness could improve the health of both these women and their babies. We found that two questions about low mood, or a 10-item self-complete questionnaire, were useful in identifying pregnant women likely to have depression or other mental illnesses. Around one in four participants had a mental illness. Women identified as being depressed were invited to participate in a study to examine whether or not a future large trial looking at the usefulness of tailored, guided self-help would be feasible. Recruitment was slow, partly because of changes in the way that services were organised, and so a larger trial would not be possible. However, the new, tailored, guided self-help materials were valued by women and practitioners.

We interviewed women who had experienced mental health problems (mild to severe) during the perinatal period (i.e. during pregnancy and the year after birth) and found that services tailored to the perinatal period were usually preferred to general services. However, stigma and fears about losing custody of children are barriers to disclosure of mental health problems.

We also developed and evaluated a new measure of service satisfaction for women with acute severe mental illnesses after birth who needed psychiatric admission or treatment by crisis resolution teams. We used this and other measures in a large trial comparing readmission rates in the year after discharge among women treated in specialist psychiatric mother and baby units and those treated in acute generic psychiatric inpatient wards or at home under the care of a crisis resolution team (i.e. standard care). Women treated in mother and baby units were slightly less likely than those who received standard care (22% vs. 25%) to be readmitted in the year after discharge. Women were also more satisfied with specialist services. However, specialist services were not good value for money compared with standard care in the short term, although this may not be the case in the longer term.

Programme Grants for Applied Research

ISSN 2050-4322 (Print)

ISSN 2050-4330 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full PGfAR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/pgfar. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Programme Grants for Applied Research journal

Reports are published in *Programme Grants for Applied Research* (PGfAR) if (1) they have resulted from work for the PGfAR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Programme Grants for Applied Research programme

The Programme Grants for Applied Research (PGfAR) programme, part of the National Institute for Health and Care Research (NIHR), was established in 2006 to fund collaborative, multidisciplinary programmes of applied research to solve health and social care challenges. Findings are expected to provide evidence that lead to clear and identifiable patient benefits, in the relatively near future.

PGfAR is researcher led and does not specify topics for research; however, the research must be in an area of priority or need for the NHS and the social care sector of the Department of Health and Social Care, with particular emphasis on health and social care areas that cause significant burden, where other research funders may not be focused, or where insufficient funding is available.

The programme is managed by the NIHR Central Commissioning Facility (CCF) with strategic input from the Programme Director. For more information about the PGfAR programme please visit the website: https://www.nihr.ac.uk/explore-nihr/funding-programmes/programme-grants-for-applied-research.htm

This report

The research reported in this issue of the journal was funded by PGfAR as project number RP-PG-1210-12002. The contractual start date was in September 2013. The final report began editorial review in June 2020 and was accepted for publication in June 2021. As the funder, the PGfAR programme agreed the research questions and study designs in advance with the investigators. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The PGfAR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health and Care Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, CCF, PGfAR or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, the PGfAR programme or the Department of Health and Social Care.

Copyright © 2022 Howard et al. This work was produced by Howard et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaption in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HSDR, PGfAR, PHR journals) and Editor-in-Chief of HSDR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Consultant in Public Health, Delta Public Health Consulting Ltd, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Adviser, Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Reader in Trials, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, Child and Adolescent Mental Health, Palliative Care and Paediatrics Unit, Population Policy and Practice Programme, UCL Great Ormond Street Institute of Child Health, London, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk